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Data Taking 
Day Delivered Recorded Efficiency Comment

(%)

24 Oct 3.07 2.73 89

25 Oct 3.45 2.82 82 Cal preamp fails.  Special runs taken.

26 Oct 0.00 0.00 - Access to repair Cal preamps.

27 Oct 4.48 3.79 85

28 Oct 2.00 1.70 85

29 Oct 3.01 2.55 85 And More Muon HV trips.

30 Oct 4.43 3.92 89

24-30 Oct 20.4 17.5 86

Lum (pb-1) Lum (pb-1)
CAL BLS PS supply dies.  Special runs 

taken.  Access to replace Supply

Three CAL towers go hot and masked off.  
Muon HV trips.

More Muon HV trips.  Access for CAL



31 October 2005                                                   AEM                                                               Bill Lee

Significant Events
● Calorimeter

– 24 October – BLS supply died.  
● This prevents us from reading out 1/12th of the calorimeter.
● Fixed in an access that evening.

– 25 October – Preamp secondary supply fails.
● The preamp supplies are located inside the detector and all have backups.  We had 

lost half of the primaries over the previous 10 months.  We then lost one of the 
secondaries.  

● We then had a 16 hour supervised access to attempt to repair all of the supplies.

– However, we still have two supplies on secondary.
– Thanks for the prompt access.

●  Muon

–  27 – 29 October – One PDT channel tripped multiple times per day. 
● Currently have raised the trip current. 
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Accesses

● 24 October ­ Controlled
– Replace Calorimeter BLS supply.

● 26 October – Supervised
– Open detector

– Repair seven preamp supplies.

– Surveyed south EMC truss.

– Replace CFT AFE PS module

– 2 old ODH heads replaced.

– Repair a few channels in A layer PDT.

● 28 October – Controlled
– Replace BLS controller card.
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High Luminosity
● Records

– Initial Luminosity – 
● Three stores in top five.
● Store 4473 set the record of 150E30.

– (Todays record of 158E30 goes on next week)

– 20.4 pb­1 is the record for delivered luminosity for a week.

– 17.5  pb­1 is second best for weekly recorded luminosity.

– D0 also has its 2nd and 3rd best days for recorded luminosity.

● Running

– D0 has started the last couple of stores without any significant problems.
– (Todays store did have some unexplained backpressure from the L3 Farm)

– Replaced L3 routing master with a higher CPU in anticipation of higher 
luminosities.
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D0 Losses

● The losses at D0 have been greatly reduced.
● Our sincere thanks go out to the AD for the progress they 

have made on this issue.

Store 4468   ­ 137E30
­ 64 R/min

Store 4477   ­ 158E30
­ 18 R/min


